Author's response to reviews

Title: Economic and other barriers to adopting recommendations to prevent childhood obesity: results of a focus group study with parents

Authors:

Kendrin R Sonneville (ksonnevi@hsph.harvard.edu)
Nancy La Pelle (Nancy.LaPelle@umassmed.edu)
Elsie M Taveras (Elsie_Taveras@harvardpilgrim.org)
Matthew W Gillman (Matthew_Gillman@harvardpilgrim.org)
Lisa A Prosser (lisapros@med.umich.edu)

Version: 4 Date: 8 December 2009

Author's response to reviews:

Uploaded draft includes corrected typographical errors. Three repeated sentences are removed and three sentences are edited with clearer language.